Author’s response to reviews

Title: Associations of BAFF rs2893321 polymorphisms with myasthenia gravis susceptibility

Authors:

Hui Deng (huideng_ny@163.com)
Jianjian Wang (wangjj_hrb@163.com)
Xiaotong Kong (xiaotongk_23@163.com)
Huixue Zhang (huixue81@163.com)
Tianfeng Wang (wg33tianfeng@163.com)
Wenqi Tian (wqtian665@163.com)
Tingting Yi (yting2g@163.com)
Lihua Wang (lihua_wlh@163.com)

Version: 4 Date: 28 Sep 2019

Author’s response to reviews:

1. Authors’ names

We note the order of the first names and last names of each author on the title page and submission system are not the same. Please amend the order of the first name and last name of each author on either the submission system or the title page of your manuscript and ensure that this order is the same on both the title page and the submission system. Once this is done, please ensure that the order of initials in the Authors’ contributions section of your manuscript are in line with this whereby the initials should be [first initial of first name][first initial of last name] i.e. Hui Deng as “HD”.

ANSWER: we have modified the authors' name in the revised MS.